Author's response to reviews

Title: A cyclopalladated complex interacts with mitochondrial membrane thiol-groups and induces the apoptotic intrinsic pathway in murine and cisplatin-resistant human tumor cells

Authors:

Fabiana A Serrano (fabianaserrano@uol.com.br)
Alisson L Matsuo (almatsu@unifesp.br)
Priscila T Monteforte (pris.farm@gmail.com)
Alexandre Bechara (alebechara@gmail.com)
Soraya S Smaili (ssmaili@unifesp.br)
Débora P Santana (deborapsantana@yahoo.com.br)
Tiago Rodrigues (tiago.rodrigues@ufabc.edu.br)
Felipe V Pereira (felipevalencap@yahoo.com.br)
Luis S Silva (silva.luiz@unifesp.br)
Joel Machado Jr (joel.machadojr@gmail.com)
Edson L Santos (edson.santos@unifesp.br)
João Bosco Pesquero (jbpesquero@unifesp.br)
Rafael M Martins (rafamiya@gmail.com)
Luiz R Travassos (luiztravassos@gmail.com)
Antonio CF Caires (caires@umc.br)

Version: 6 Date: 11 July 2011

Author's response to reviews: see over
Dear Editor,

We presently submit the FINAL REVISION of the manuscript

MS: 6010538244617825

All major changes to conform the journal style are highlighted.
Figures and Tables numbering was corrected.

We hope to hear from you soon.

All the best,

Elaine Guadalupe Rodrigues, Ph.D.
Professor
Experimental Oncology Unit
Department of Microbiology, Immunology and Parasitology
Federal University of São Paulo
Brazil